Author's response to reviews

Title: Validity of gout diagnosis in Swedish primary and secondary care

Authors:

Mats Dehlin (mats.dehlin@vgregion.se)
Kalliopi Stasinopoulou (kalliopi.stasinopoulou@vgregion.se)
Lennart Jacobsson (lennart.jacobsson@gu.se)

Version: 4 Date: 27 December 2014

Author's response to reviews: see over
To the editor of BMC Musculoskeletal Disorders

Revisions made in the article “Validity of gout diagnosis in Swedish primary and secondary care” for publication in BMC Musculoskeletal Disorders.

Email addresses of all authors are included in the title page of the manuscript.

Heading of the abstract is included in the manuscript.

Line and page numbers have been included in the manuscript.

Regards

Mats Dehlin, corresponding author

Department of rheumatology and inflammation research,
P.O. Box 480,
405 30, Gothenburg

Sweden

e-mail: mats.dehlin@vgregion.se